Author's response to reviews

Title: Hypertrophy of the feet and ankles presenting in a primary hypertrophic osteoarthropathy or Pachydermoperiostosis : a case report

Authors:

   Rim Akrout (akroutrim@yahoo.fr)
   Samar BenDjemaa (bendjima@gmail.com)
   Héla Fourati (hela.fourati@tunet.tn)
   Mariem Ezzeddine (mariam.ezzeddine@laposte.net)
   Imene Hachicha (mainouhachicha@yahoo.fr)
   Soufiene Baklouti (sofien.baklouti@rms.tn)

Version: 5 Date: 16 November 2011

Author's response to reviews:

I have modified the manuscript. The title is like your request. the dates were modified and in the figure 1, the patient’s eyes were removed.